Author's response to reviews

Title: Public awareness of major infectious diseases in a rural population of China: a cross-section study

Authors:

He Liu (annliuhe@163.com)
Mei Li (missmeili@hotmail.com)
Mingjuan Jin (jinmi@zju.edu.cn)
Fangyuan Jing (jingfy123@yahoo.com.cn)
Hui Wang (xiaoxiaohuiya@163.com)
Kun Chen (ck@zju.edu.cn)

Version: 2 Date: 11 June 2012

Author's response to reviews: see over
Dear editorial board of *BMC Infectious Diseases*:
We would like to submit the enclosed manuscript entitled “Public awareness of major infectious diseases in a rural population of China: a cross-section study”, which we wish to be considered for publication in *BMC Infectious Diseases*. No conflict of interest exits in the submission of this manuscript, and manuscript is approved by all authors for publication. I would like to declare on behalf of my co-authors that the work described was original research that has not been published previously, and not under consideration for publication elsewhere, in whole or in part. All the authors listed have approved the manuscript that is enclosed.
In this work, we evaluated the level of awareness of HIV, HBV and TB as well as associated factors among rural residents in China. The study based on the Mega-projects of Science Research for the 11th and 12th Five-Year Plan of China, which supplied enough human, material and financial resources to finish the field survey of big sample. I hope this paper is suitable for *BMC Infectious Diseases*.
We deeply appreciate your consideration of our manuscript, and we look forward to receiving comments from the reviewers. If you have any queries, please don’t hesitate to contact me at the address below.
Thank you and best regards.
Yours sincerely,

Corresponding author:
Name: Kun Chen
E-mail: ck@zju.edu.cn